Contents lists available at ScienceDirect



European research on management and business economics

journal homepage: www.elsevier.es/ermbe



Board of directors and firm debt in Spanish SMEs: A power perspective



Juan Francisco Martín-Ugedo, Antonio Mínguez-Vera

Faculty of Economics and Business, Department of Management and Finance, University of Murcia, Campus de Espinardo, s/n, Murcia 30100, Spain

ARTICLE INFO

JEL-Classification: G30 G32 Keywords: Board of directors Power Firm debt This manuscript examines the influence of some variables relating to boards of directors that we consider to be proxies for power (gender diversity, duality, board size and insider ownership). Their influence on firm debt is explored. The sample examined, Spanish non-financial SMEs, has some particular characteristics. Share ownership, which is usually highly concentrated in the Civil Law context, is even more concentrated when we focus on SMEs. As a consequence, there is little separation between ownership and control. We have employed a power perspective as the main theoretical framework. Data were obtained from the SABI database, and the methodology employed is a panel data, applying the System GMM technique. This methodology makes it possible to control for endogeneity and individual heterogeneity. The results show that a larger proportion of female directors, larger boards, the separation of the roles of CEO and Chairman, and more shares owned by directors and by the CEO stimulate a decrease in firm debt. This evidence supports the hypotheses tested.

1. Introduction

There are many studies of the effect of corporate governance variables on firm outcomes (Valls Martínez et al., 2022). Many such studies have focused on the effect on firm performance and firm returns (García-Martín and Herrero, 2018), but there have been fewer studies of the effect on risk.

Corporate governance mechanisms of control are intended to align the interests of owners and executives of companies. These mechanisms can be external or internal to the business. Spain is considered a civil law country. In contrast, the American market, where most previous evidence comes from, is classified as a common law country. Countries with civil law systems have less investor protection and more concentrated ownership structures. Thus, internal control mechanisms, including the board of directors, are stronger (La Porta et al., 2002). As a consequence, board composition may be a more effective way to align the interests of shareholders and executives in these countries.

Most studies of corporate governance employ agency theory, or more specifically the principal-agent perspective. However, in Spain, as a Civil Law country, the conflict usually occurs between majority and minority shareholders (tunneling effect, Johnson et al., 2000). Although the majority shareholders in Spain are motivated to monitor the actions of executives, they themselves also usually serve as executive directors, thereby gaining the opportunity to negotiate agreements with the other members of the board, making decisions that will benefit themselves regardless of the interests of the minority shareholders (Sánchez-Marín

et al., 2022).

This study adopts the power perspective as its main theoretical framework. We argue that some characteristics of the board (and the CEO) may be considered proxies for power, and that the firm will assume more financial risk in proportion to the power of the board (and the CEO).

Because of the possibility of bankruptcy and financial distress, and the cost that these situations provoke (Baxter, 1967; Warner, 1977), we consider that firm debt is a proxy for firm (financial) risk. Thus, we hypothesize that companies in which the power of the board of directors is highly concentrated in a few people will incur higher levels of firm debt. The power variables that we employ are: Gender diversity (proportion of women directors), duality (the fact that different people assume the roles of chairman of the board and CEO), the size of the board, and the percentage of shares owned by the board and the CEO.

This manuscript is important as it makes several contributions to the literature. Many studies of corporate governance and corporate finance focus on the effect of some board characteristics on firm value, firm performance, firm returns and similar variables. In contrast, this manuscript focuses on the effect on a proxy for firm financial risk, namely, the level of debt of the firm. Most evidence that has been related to similar topics has been collected about large firms and not SMEs. Given the characteristics of the sample examined, SMEs, we have decided to adopt a power perspective as the main theoretical approach. As far as we know, no previous study has followed this approach for a similar analysis. These two particularities, the focus on firms' debt and

https://doi.org/10.1016/j.iedeen.2023.100231

Received 13 June 2023; Received in revised form 20 October 2023; Accepted 25 October 2023 Available online 2 November 2023 2444-8834/@ 2023 The Author(s) Published by Elsevier España S L U on behalf of AEDEM

^{*} Corresponding author. *E-mail address:* minver@um.es (A. Mínguez-Vera).

^{2444-8834/© 2023} The Author(s). Published by Elsevier España, S.L.U. on behalf of AEDEM. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

J.F. Martín-Ugedo and A. Mínguez-Vera

the theoretical approach employed, fill a gap in the literature. In addition, from the methodological point of view, we employ a panel data methodology, applying the System GMM technique. This is a powerful methodology that makes it possible to control for endogeneity, individual heterogeneity and autocorrelation.

The results show, first of all, a negative effect on indebtedness of the presence of women on boards of directors, the separation of positions of chairman of the board and CEO and the size of the board. However, when directors and CEO own more shares, corporate debt increases. The interrelation between the percentage of women on the board and the share ownership of the directors and CEO reduces the company's debt. This does not occur when the separation of positions and board size are combined with ownership variables. Finally, it is not found that the interrelation between the presence of female directors and the separation of positions, on the one hand, and board size, on the other, have a positive effect on corporate debt.

The rest of the manuscript is structured as follows. Section 2 presents the theory and the hypotheses. Section 3 describes the data and the methodology. Section 4 presents the results. Finally, Section 5 presents the conclusions.

2. Theory and hypotheses

Board of directors is the main body that exercises corporate control on behalf of shareholders. Board effectiveness depends on many variables, such as its composition, size, and so on (John & Senbet, 1998; Mínguez-Vera & Martín-Ugedo, 2010).

Several theories, such as agency theory, resource dependency and stewardship have been used to explain the development of the board of directors (de Enrique Arnau & Pinillos-Costa, 2023) but, as Banerjee et al. (2020) show, agency is the most frequently employed.

There are at least two reasons why literature pays most attention to the board. First, the great responsibility adopted by this body and, second, the fact that shareholders are demanding more of board members. Board members are assuming the more and more responsibilities and risks that were traditionally considered to be the CEO's, who was held to account in case of corporate difficulty (Banerjee et al., 2020).

De Enrique Arnau and Pinillos-Costa (2023) observe the that literature on boards has changed its focus. Until 2007 the interrelations between board members and CEO and the roles of institutional investors where prevalent. But in more recent studies attention has moved to societal issues, such as sustainability, environmental aspects and diversity.

The risk accepted by the firm is among the responsibilities of the board. Risk is usually associated with the difficulty of estimating the exact value of a parameter. Measures of statistical dispersion are frequently employed as proxies for risk. Some characteristics of the board may be determinants of firm risk. The literature on group decision-making and the concept of power provide arguments to explain this relationship. Power is usually considered to be the ability to overcome resistance to attain a goal (Pfeffer, 1981). Thus, power is directly related to the possibility of influencing decisions. As a consequence, factors such as power distribution, which depends on variables such as managerial discretion, participation, and demographic and cultural diversity are central to this relationship.

Group members can disagree. The decisions adopted by a group require some sort of consensus. The final decision of a group involves differing opinions, representing an average, a compromise among individual positions (Moscovici & Zavalloni, 1969). As a consequence, decisions adopted by individuals are likely to be riskier than decisions adopted by a group (Halliday et al., 2021).

However, as previously mentioned, it is not only the size of the group that matters, but also the distribution of power among its members. When power is unevenly distributed, the opinions of the leading members, as well their errors of judgment, cannot be balanced by other members' opinions, making it more likely that the group adopts riskier decisions.

Capital structure was a recurrent topic in the financial literature, until Modigliani and Miller (1958) demonstrated that the leverage adopted by a firm is irrelevant in perfect capital markets. After that, the literature focused on different market imperfections such as taxation, financial distress, agency costs and information asymmetry. As a consequence, there are two main, non-exclusive, theories to explain firms' capital structures, the trade-off theory and pecking order theory (Myers & Majluf, 1984). Both theories take into consideration the risk level that underlies decisions about capital structure.

The trade-off theory considers that a firm has an optimal debt level that is reached by balancing the advantages and disadvantages of market imperfections. One of those disadvantages is the cost associated with financial distress.

The literature on capital structure states that high levels of debt lead to higher (financial) risk and to the possibility of financial distress (Baxter, 1967; Warner, 1977; Fasano et al., 2022). Therefore, we hypothesize that some governance variables, which may be considered proxies for power, may exert an influence on the debt level of firms. According to Sánchez-Vidal et al. (2020), corporate debt can cause negative effects such as reduced flexibility when obtaining future financing, lower growth and more job losses in the event of financial crises.

To develop the empirical analysis in this study, we focus on Spanish SMEs. According to the International Council of Small Business (ICSB), 2023, SMEs represent 90 % of the business sector worldwide, generate between 60 % and 70 % of employment and are key to achieving the Sustainable Development Goals (SDGs).

SMEs are the engine of the Spanish economy. According to the latest data from Ministerio de Industria, Comercio y Turismo (2023), they represent 99.8 % of Spanish firms and generate 66.4 % of employment. Furthermore, in the last economic crisis, they began a process of internationalization, seeking new markets, because of the weakness of domestic demand. According to Eurostat (2023) data, they are responsible for 53.3 % of Spain's imports and 51.1 % of the exports.

In Europe, small and medium-sized companies are especially important in job creation and they generate half of the intra-community trade in goods. They account for 51 % of the imports and 45 % of the exports within the EU.

Most previous evidence on corporate governance focuses on large and/or quoted firms. However, in recent years, interest in the boards and CEO of SMEs has increased. The boards of SMEs tend to be less formalized, less structurally complex, and more homogeneous (Forbes & Milliken, 1999). As a consequence, the range and depth of tasks developed by each member are more diverse, and the possibility of influencing decisions is greater (Judge & Zeithaml, 1992). In addition, the directors of SMEs usually have less experience, knowledge and skills, reducing the effectiveness of the board (Gorman & Sahlman, 1989).

On the other hand, share ownership, which is usually highly concentrated in the Civil Law context, is even more concentrated in SMEs. In addition, shareholders and management are frequently the same people, reducing the possibility of exercising effective control (Forbes & Milliken, 1999).

Next, we present the arguments and hypotheses relating to the influence that some variables that we consider proxies for power may exert on risk and, as a consequence, on firm debt. These are the proportion of women directors, CEO duality, the size of the board, and the percentage of shares owned by the board and the CEO.

2.1. Gender diversity (proportion of women directors)

Many different theoretical perspectives examine the relationship between board gender diversity and other firm variables (Valls Martínez et al., 2022). Terjesen et al. (2009) conducted a survey, and identified about twenty different theories on gender diversity and corporate governance. These theories included agency theory, institutional theory,

J.F. Martín-Ugedo and A. Mínguez-Vera

resource dependency, human capital theory, and a range of social psychological theories. These theories usually overlap, meaning that most studies do not consider a single theory or approach.

From an institutional point of view, there are country-level differences in female representation on boards of directors. National culture, the legal factors and systems, and corporate ownership structures are among the most important factors determining such representation (Grosvold & Brammer, 2011; Halliday et al., 2021). Spain, as a civil law country, has high ownership concentration and its national culture has a relatively low level of masculinity (implying a disposition to seek consensus), especially in comparison with Anglo-Saxon countries, and a high level of uncertainty avoidance (Hofstede, 2021; La Porta et al., 1999).

Different arguments point to a reduced level of firm financial risk when there is a larger proportion of women directors (López-Delgado & Diéguez-Soto, 2020). Women seem to be more risk averse that men. This evidence comes not only from the finance and strategy fields, but also from fields such as professional work in health and physical safety (Maxfield et al., 2010). In addition, it is important to consider how people interact in a group, and therefore the importance of gender diversity.

Several theories of social groups examine how people with similar/ different values and profiles interact. These theories include social network theory and social cohesion theories, as well as theories of ingratiation and social identity. Social identity theory (Tajfel & Turner, 2004) posits that individuals divide members of their organization into members of their group (in-groups), or individuals similar to themselves, and members from outside their group (out-groups), or individuals different from themselves. They tend to perceive the former positively, and the latter negatively (Nielsen & Huse, 2010). As a consequence, individuals seek to surround themselves with people who share similar profiles. Homogeneous groups tend to adopt riskier decisions (Roberson, 2019; Watson et al., 1993). Thus, having more women directors should lead to less risky decisions.

There is much previous evidence that shows that women are more risk averse that men (Chaganti, 1986; Khan & Vieito, 2013). This risk aversion on the part of women has also been observed in the arena examined in this manuscript, as Carter and Shaw (2006), among others, have found lower debt levels in firms managed by women. García and Herrero (2021) found that gender diversity of the board is negatively related to leverage and the cost of debt.

Related to agency theory, Carter et al. (2003) suggest that more diverse boards increase board independence. This is why increasing gender diversity could improve monitoring and control of management. They comment that women are more inclined to ask questions that would not be asked by male directors. Supporting these arguments, Adams and Ferreira (2009) found evidence suggesting that female directors provide greater insight and closer monitoring. This circumstance can increase risky decisions because the board of directors will be more aligned with the interests of external shareholders, who are less risk averse than insiders.

On the other hand, numerous studies affirm that women exercise power differently. Kelly et al. (2008) suggest that men use more domination and intimidation, while women tend to be more affectionate. Female leaders are seen to adopt a more democratic and participative style than their male counterparts and prefer consensus (Merchant, 2012). Rosener (1990) argues that female leadership tends to encourage participation, promote sharing power and information, favor general good over self-interest, and relate power to interpersonal skills. As a result, women may be more participatory in decision-making, which could mean less risky decisions.

In line with most of the previous arguments, we propose the following hypothesis:

Hypothesis 1. A larger proportion of women on the board will lead to less risky decisions, and thus lower levels of debt.

2.2. Duality

CEO duality, CEO-Chairman duality, or just duality, refer to the practice of different persons holding the positions of chairman of the board and CEO. It is one of the most frequently recurring topics in the literature on corporate governance.

There are two main theories that are used to explain the effect of duality-non duality on firm performance: agency theory and stewardship theory. Both theories agree that duality implies higher board control and weaker CEO power (Finkelstein, 1992). Moreover, appointing one person to both posts sends a clear message of leadership (Donaldson & Davis, 1991). This could skew the distribution of power.

Studies focusing on the influence of duality or non-duality on risk are scarce. Agency theory points to a conflict, in terms of the optimal level of firm risk, between the external shareholders and insiders. External shareholders can diversify their investments, reducing their risk, but insiders suffer the consequences of firm risk taking more directly (Jensen & Meckling, 1976). Combining the posts of CEO and chairman reduce supervision of the insiders and this can reduce firm risk. Daily and Dalton (1994a, 1994b) examined the relationship between non-duality and firm bankruptcy. They hypothesized that non-duality would imply a larger concentration of power, and that this would increase the likelihood of bankruptcy. The results are mixed as Daily and Dalton (1994a) found that duality increased the likelihood of bankruptcy, but Daily and Dalton (1994b) found no significant effect.

In contrast with this, the power perspective assumes that duality increases the power of the person who holds both posts, making it possible for them to take more risk on behalf of the firm (Halliday et al., 2021; Mínguez-Vera & Martín-Ugedo, 2010).

According to the previous arguments, we propose the following hypothesis:

Hypothesis 2. If a single person holds both positions (CEO and chairman), this will lead to riskier decisions being adopted and thus higher levels of debt.

2.3. Board size

Size is another variable that determines board effectiveness in its monitoring function. Even if managers have the right incentives, they can make mistakes (Sah & Stiglitz, 1986, 1991). Given that group members can disagree, group decisions imply differing opinions, requiring some sort of consensus on decisions. There should be less risk in decisions made by group consensus than in decisions made by individuals.

Sah and Stiglitz (1986) developed a model in which the larger the group of decision-makers, the less likely they were to accept a bad investment project. And Sah and Stiglitz (1991) found that highly variable returns, implying higher levels of risk, are less likely in firms with many managers. The larger the size of the decision group, and less power for individual decision-makers, reduce firm risk (Mínguez-Vera & Martí-n-Ugedo, 2010).

Large boards may benefit from a greater variety of opinions and experiences and increased monitoring capacity. In contrast, coordination/communication problems are also more severe (Hermalin & Weisbach, 2003). These problems of communication and decline of individual responsibility are more pronounced in larger boards because the cost to any director of not exerting due diligence in monitoring falls in proportion to the total number of board members.

Considering these arguments, Yermack (1996) and Core et al. (1999) suggest a curvilinear relationship between board size and monitoring effectiveness, according to agency theory. In small boards, as the number of directors grows, supervision of top executives increase, resulting in an alignment of interests and riskier decisions. However, when an optimum number of directors is reached, the trend reverses. Jensen (1993) argues that when a board has more than seven or eight directors,

it becomes more difficult to exert effective control over the CEO. So, the final effect it is not clear, according to this theory.

In addition, as previously mentioned, according to the power perspective, the final decision of a group represents an average among individual positions. Thus, decisions adopted by a group are, on average, less risky than those adopted by individuals (Moscovici & Zavalloni, 1969; Sah & Stiglitz, 1986). As a consequence, increasing the size of the board should reduce firm risk (Halliday et al., 2021).

Previous evidence on board size focuses mainly on its effect on firm value, and not on risk. However, most evidence indicates that board size has a negative impact on a firm's risk (Cheng, 2008; Nakano & Nguyen, 2012).

According to the power perspective, we propose the following hypothesis:

Hypothesis 3. A larger board will lead to less risky decisions adopted and thus lower levels of debt.

2.4. Insider ownership (Percentage of shares owned by the board and the CEO)

As in the case with the board size, the effect of insider ownership is not clear in the context of agency theory. Morck et al. (1988) argue that low levels of director ownership lead to more effective control (convergence hypothesis). This alignment of interests between shareholders and insiders can lead to riskier decisions being made (Mínguez-Vera & Martín-Ugedo, 2010). However, high levels of director ownership may lead directors to entrench themselves (entrenchment hypothesis) with the opposite effect.

In general, a large concentration of shares in the hands of a small number of shareholders will give those shareholders more power, and this power is increased when those shareholders are insiders. As a consequence, it is expected that when insiders have large holdings of shares they will be able to decide that the firm will take more risk, and incur more debt. In addition, large shareholders may be reluctant to engage in equity financing, as that would tend to dilute their ownership. To maintain the control of the firm, they prefer debt financing.

Unlike the American market, where more dispersed ownership structures have been observed, Spanish SMEs have highly concentrated ownership structures (La Porta et al., 1999; Leech & Manjón, 2002). Spanish companies are mainly controlled by internal, dominant shareholders, who frequently act simultaneously as directors and managers (Baixauli-Soler & Sanchez-Marin, 2015). In this scenario, controlling shareholders exercise considerable power.

According to previous arguments, we consider that larger insider ownership will lead to more concentrated power and thus we present the following hypothesis:

Hypothesis 4. More insider ownership will lead to riskier decisions adopted and thus higher levels of firm debt.

3. Sample and methodology

The sample includes Spanish non-financial SMEs, between 2014 and 2019, obtained from the SABI database. We use the size criterion proposed by the European Commission (2003), according to which SMEs employ fewer than 250 people and have either an annual turnover not exceeding \notin 50 million, or an annual total balance sheet not exceeding \notin 43 million. After selecting SMEs from the database, removing companies without data and with errors, we have an unbalanced panel of 2385 companies and 9715 observations.

We consider the firm's leverage (LEVER), computed as the ratio of total debt to total assets, as the dependent variable (Hernández-Cánovas et al., 2016; Hernández-Nicolás et al., 2019). Alternatively, and for robustness, the analyses have been repeated including the long-term and short-term debt ratio as dependent variables. The results do not differ significantly from those shown for total debt. The results for those two

additional analyses have not been included for synthesis reasons. As independent variables, we measure the gender diversity or the proportion of women on the board, PWOMEN, as the number of female directors divided by the total number of directors (Valls Martínez et al., 2022). For duality, a dummy variable, DUAL, takes a value of 0 when the CEO and chairman of the board are the same person and 1 otherwise (Mishra & Mohanty, 2014). The board size, LNDIR, is computed as the logarithm of the number of directors on the board (Cashman et al., 2012). And insider ownership is calculated as the proportion of shares owned by the directors and CEO, DIROWN and CEOWN respectively (Fahlenbrach et al., 2010). We also include the squares of these two variables, DIROWN2 and CEOWN2, in order to account for a possible non-monotonic effect.

Finally, we consider the heterogeneity across firms by including several firm-specific control variables in the model. Variables generally used as determinants of leverage are (Brav, 2009; Gaud et al., 2005; Hall et al., 2004) the firm's size measured as the logarithm of total assets, FSIZE; the firm's age, approximated by the logarithm of the number of years the firm has been established, LFAGE; the ratio of intangible assets to total assets, INTAS; the debt cost, DEBCOS, measured as the ratio of financial expenses to total debt; the return on assets, ROA, calculated as the ratio of profits, before financial expenses and taxes, to total assets; and the interest coverage ratio, INCO, the number of times that net profit before interest and taxes related to a firm can cover interest payments.

Table 1 shows the descriptive statistics of all variables. The mean value of the proportion of women on the board is only 7.2 %. We also observe that only a third of the firms in our sample have a separate power structure relating to CEO and Chairman, although the Spanish Good Governance Code (2020) recommends the separation of those posts. The number of directors, NDIR, has a mean value of 5, just in the lower level of the recommendation of the Spanish Good Governance Code for listed firms: "*The board of directors should have an optimal size to promote its efficient functioning and maximize participation. The recommended range is accordingly between five and fifteen members*". Finally, the ownership of directors has a value of 35.2% while the percentage of shares owned by the CEO is about 18.4 %. These values are higher than those reported by Mínguez-Vera and Martín-Ugedo (2010) for Spanish listed firm (3 % and 2 %, respectively).

According to the theoretical arguments, the following model is proposed to capture the relationship between the characteristics of the board of directors and the debt ratio of the firm:

Table 1	
Descriptive	Statistics.

Mean	Median	Standard Deviation
0.630	0.659	0.209
0.072	0.000	0.150
0.335	0.000	0.472
4.957	4.000	3.806
1.329	1.386	3.806
0.352	0.386	0.200
0.184	0.285	0.500
16.235	16.173	0.826
2.908	2.933	0.598
0.219	0.174	0.186
0.029	0.025	0.035
0.084	0.068	0.082
1.966	0.313	30.747
	0.630 0.072 0.335 4.957 1.329 0.352 0.184 16.235 2.908 0.219 0.029 0.084	0.630 0.659 0.072 0.000 0.335 0.000 4.957 4.000 1.329 1.386 0.352 0.386 0.184 0.285 16.235 16.173 2.908 2.933 0.219 0.174 0.029 0.025 0.084 0.068

VARIABLES: LEVER (leverage ratio); PWOMEN (proportion of female directors on the board); DUAL (dummy variable that takes a value of 1 when the chairman of the board and CEO are not the same person, and zero otherwise); NDIR (board size); LNDIR (the logarithm of the number of directors on the board); DIROWN (proportion of shares owned by the directors); CEOWN (proportion of shares owned by the CEO); FSIZE (firm size, measured as the logarithm of total assets); LFAGE (logarithm of the firm age in years); INTAS (ratio of intangible assets to total assets); DEBCOS (debt cost, measured as the ratio of financial expenses to total debt); ROA (return on assets); INCO (interest coverage ratio).

$$\text{LEVER}_{it} = \beta_0 + \beta_1 \text{BOARD}_{it} + \sum_{j=2}^n \beta_j \text{OTHER}_{jit} + \psi_t + \eta_i + \varepsilon_{it}$$

where LEVER refers to the debt ratio; BOARD includes the different characteristics of the board of directors (the proportion of female directors, PWOMEN; leadership structure, DUAL; the board size, LNDIR; and the proportion of shares owned by the board members, DIROWN and the CEO, CEOWN). OTHER refers to the control variables (firm size, FSIZE; the logarithm of the firm age, LAGE; the intangible assets to total assets ratio, INTAS; the debt cost; the return on assets, ROA; and the interest coverage ratio, INCO). Finally, ψ_t , η_i and ε_i are the time effects, the unobserved individual heterogeneity and the error term, respectively.

Considering the hypotheses presented in Section 2, a negative sign is expected for the proportion of female directors, PWOMEN, the dummy that indicate the separation of the posts of CEO and Chairman, DUAL and for the board size, LNDIR, on debt ratio, LEVER. According to the power perspective, we expect a positive relationship between the proportion of shares owned by board members, DIROWN and the CEO, CEOWN and firm debt.

For the control variables, there are several arguments why firm size is related to capital structure (Kieschnick & Moussawi, 2018), including economies of scale in lowering information asymmetry, and scale in transaction costs, and market access. These reasons suggest that smaller firms should have lower debt levels. So a positive sign between firm size and debt ratio is expected.

A negative relationship between the age of the firm, LAGE and the leverage level is expected. Older firms have accumulated funds and, therefore, will probably require fewer external funds (Michaelas et al., 1999). With regard to the intangible assets to total assets, INTAS, a negative sign can be obtained as a consequence of the indirect correlation between the level of intangible assets and the debt structure of the firm (Parsons & Titman, 2009). The expected sign for the debt cost, DEBCOS is negative (firms with higher cost of debt will have a lower debt ratio) (Barry et al., 2008).

For Return on Assets, ROA, Modigliani and Miller (1963), argue that tax shielding would induce profitable companies to use more debt. In contrast, the pecking order theory (Myers & Majluf, 1984) concludes that profitable firms would primarily finance with retained earnings rather than external financing. Finally, the interest coverage ratio, INCO (Robinson et al., 2015), shows the ability of a firm to repay its debt (Zarebski & Dimovski, 2012) and it is a measure of the riskiness of a firm (Andrade & Kaplan, 1998). Risk of insolvency is higher when earnings are relatively low with respect to interest expenses. Consequently, a positive relationship is expected.

Equations are estimated using a panel data methodology, applying the System GMM technique (Arellano & Bover, 1995; Blundell and Bond, 1998). This methodology makes it possible to control for individual heterogeneity (Himmelberg et al., 1999). These are unobserved variables that affect each of the firms in the sample unequally, but which are constant over time and which directly affect the decisions made by these units. Descriptions of the kind of things usually considered under this heading are entrepreneurial capacity, operational efficiency, capitalization of experience, and access to technology. The methodology also controls for macroeconomic effects on the dependent variable (those factors that equally affect all the companies in the study). In addition, System GMM estimation solves the endogeneity problem that may appear when the independent variables and the dependent variable are determined simultaneously, as is the case in the present study (for example, women can choose to serve on board of firms with less debt because of their risk aversion). In order to remove any possible bias arising from simultaneous estimation, this methodology estimates a system of two simultaneous equations. The first equation uses variables in levels (first differences instruments) and the other uses variables in first differences (lagged with respect to instruments).

Specification tests proposed by Arellano and Bover (1995) are applied to examine the validity of the instruments in System GMM estimation. First, the Arellano–Bond test for the serial correlation is adopted to test whether there is a second-order serial correlation in the first-differenced residuals. The null hypothesis is that the residuals are serially uncorrelated. If the null hypothesis cannot be rejected, it provides evidence that there is no second-order serial correlation and the System GMM estimator is consistent. Second, the Sargan test is applied to test the null hypothesis of instrument validity and the validity of the additional moment restriction necessary for system GMM, respectively. Failure to reject this null hypothesis means that the instruments are valid.

This methodology has important advantages over others. The OLS (Ordinary Least Squares) estimations do not consider the unobservable heterogeneity of firms, and considers the variables to be exogenous. Fixed Effects estimation addresses the problem of unobserved heterogeneity, but not the endogeneity. The 2SLS (Two Stage Least Squares) considers the endogeneity, but it is not efficient in samples with a large number of firms and small number of years and it also has the problem of the choice of instruments (Arellano & Bond, 1991).

4. Results and discussion

The results are shown in Tables 2 to 5. In Model 1 of Table 2, we find a negative and significant effect of the proportion of female directors (PWOMEN) on firm debt ratio (LEVER). This finding supports Hypothesis 1 (A larger proportion of women on the board will lead to lower levels of debt) and is consistent with arguments from the power perspective and social identity theory (Tajfel & Turner, 2004; Roberson, 2019; and Watson et al., 1993). According to these authors, heterogeneous groups make less risky decisions. According to these arguments, decision making will be more complicated in a heterogeneous group. To decide, a consensus must be reached and the power of the male directors (with a riskier profile) can be diluted by the preferences of the female directors. Olsen and Cox (2001) found that women investors weight risk attributes, such as the possibility of loss, more than men. The World Bank (2012) report shows that women tend to be more cautious and have a lower level of ambition. Carter and Shaw (2006) also found that the presence of women in management is associated with lower levels of debt.

This result is important since it shows how female directors influence financial decisions and the survival of the company, and it suggests an important contribution to the empirical findings.

Model 2 shows a negative relationship between the separation of responsibilities (DUAL) of CEO and Chairman and firm debt. This result supports Hypothesis 2 (*If a single person holds both positions (CEO and chairman), this will lead to higher levels of debt)*. So, according to the power perspective, when the post of CEO and Chairman coincide in the same person the decision maker can accumulate more power, and so the firm will have a higher proportion of debt. This evidence contradicts the conclusion of Adams et al. (2005) who found no significant relationship between firm risk and duality. However, Daily and Dalton (1994a) found that duality increased the likelihood of bankruptcy. The findings of this research contribute to reducing the controversy surrounding CEO duality in leadership by furnishing empirical evidence of how CEO duality in corporate governance structures affects managerial behavior in corporate strategic management.

Finally, in Model 3, we observe a negative influence of board size (LNDIR) on firm leverage (LEVER) as anticipated in Hypothesis 3 (*A larger board will lead to lower levels of debt*). So, as the power perspective predicts, increasing the size of the decision-making group reduces the power of a specific decision-maker and, as a consequence, the firm risk.

In relation to the control variables, Table 2 shows a positive effect of firm size (FSIZE) on firm debt (LEVERAGE), in line with our expectations and the arguments of Kieschnick and Moussawi (2018) and Titman and Wessels (1988). The signs of the firm age, LFAGE, proportion of intangible assets, INTAS and debt cost, DEBCOS, are all as expected, and

Table 2

System GMM Estimation of the influence of female directors (PWOMEN), CEO duality (DUAL) and board size (LNDIR) on leverage ratio (LEVER).

,		0	-
Variable	Model 1	Model 2	Model 3
Constant	0.352***	0.155***	0.467***
	(0.0571)	(0.0688)	(0.0618)
PWOMEN	-0.080***		
	(0.0103)		
DUAL		-0.066***	
		(0.0058)	
LNDIR			-0.029***
			(0.0023)
FSIZE	0.037***	0.051***	0.030***
	(0.0036)	(0.0044)	(0.0039)
LFAGE	-0.081***	-0.085***	-0.073***
	(0.0012)	(0.0014	(0.0014)
INTAS	-0.194***	-0.193***	-0.166***
	(0.0186)	(0.0216)	(0.0211)
DEBCOS	-0.546***	-0.499***	-0.771***
	(0.0580)	(0.0955)	(0.0915)
ROA	-0.228***	-0.276***	-0.113**
	(0.0417)	(0.0566)	(0.0536)
INCO	0.001***	0.001***	0.001***
	(0.0021)	(0.0003)	(0.0002)
Year dummies	Yes	Yes	Yes
F	514.87***	387.10***	424.06***
z ₁	0.000	0.000	0.000
z ₂	0.000	0.000	0.000
m2	1.64	1.08	1.65
Sargan	0.183	0.151	0.147

*, **, *** Significant at 10 %, 5 % and 1 %, respectively. Standard errors in brackets.

VARIABLES: LEVER (leverage ratio); PWOMEN (percentage of female directors on the board); DUAL (dummy variable that takes a value of 1 when the chairman of the board and CEO are not the same person, and zero otherwise); LNDIR (the logarithm of the number of directors on the board); FSIZE (firm size); LFAGE (logarithm of the firm age); INTAS (intangible assets to total assets); DEBCOS (debt cost); ROA (return on assets); INCO (interest coverage ratio); F statistic (test of combined significance); Hausman (Hausman specification test), z1 and z₂, are two Wald tests of the joint significance of the reported coefficients and the joint significance of the time dummy variables, respectively (asymptotically distributed as λ^2 under the null hypothesis of no relationship, probability is shown); m₂ is a second-order serial correlation test using residuals in first differences, asymptotically distributed as N(0,1) under the null hypothesis of no serial correlation; Sargan is a test of the over-identifying restrictions, asymptotically distributed as λ^2 under the null hypothesis of no correlation between the instruments and the error term, degrees of freedom in parentheses, probability is shown.

consistent with the arguments of Michaelas et al. (1999), Parsons and Titman (2009) and Barry et al. (2008), respectively. Supporting the pecking order theory (Myers & Majluf, 1984), we find that profitable firms tend to use less debt. Finally, firms with less risk, as indicated by the interest coverage ratio, INCO, have more debt in proportion (Andrade & Kaplan 1998).

Table 3 shows the effect of directors and CEO ownership, DIROWN and CEOWN, on firm leverage (LEVER).

In Model 1, we find a positive and significant influence of the directors' ownership on firm debt, in agreement Hypothesis 4 (*More insider ownership will lead to higher levels of firm debt*). It is possible that high share ownership of directors gives them more power, and thus makes the firm more likely to adopt risky policies. A similar result is observed in Model 3, focusing on CEOs' ownership, supporting Hypothesis 4. We do not find a quadratic relationship (Models 2 and 4).

This result is in the line with that of Palmer and Wiseman (1999). They showed that the stockholdings of senior management have a positive impact on managerial risk taking. Bajaj et al. (1998) suggest that in firms with low executive ownership, directors are incentivized to under-lever the firm to reduce bankruptcy risk. May (1995) also argues that CEOs with high ownership will display riskier behavior, such as selecting investments with more risk. We contribute by adding new

Table 3

System GMM Estimation of the influence of the percentage of shares owned by directors (DIROWN) and the percentage of shares owned by CEO (CEOWN) on leverage ratio (LEVER).

Variable	Model 1	Model 2	Model 3	Model 4
Constant	0.763***	-0.209**	0.040	0.051
	(0.0816)	(0.0874)	(0.0319)	(0.1233)
DIROWN	0.001***	0.001**		
	(0.0001)	(0.0006)		
DIROWN2		-0.0006		
		(0.0006)		
CEOWN			0.001***	0.001***
			(0.00005)	(0.0005)
CEOWN2				-0.00001
				(0.000007)
FSIZE	0.013**	0.069***	0.053***	0.053***
	(0.0050)	(0.0054)	(0.0020)	(0.0077)
LFAGE	-0.100***	-0.083***	-0.080***	-0.078***
	(0.0019)	(0.0066)	(0.0008)	(0.0066)
INTAS	-0.065***	-0.104***	-0.117	-0.156***
	(0.0234)	(0.0174)	(0.0118)	(0.0400)
DEBCOS	-0.490***	-0.550***	-0.495***	-0.554***
	(0.0502)	(0.0476)	(0.0261)	(0.0650)
ROA	-0.479***	-0.009	-0.037***	-0.018
	(0.0588)	(0.0269)	(0.0137)	(0.0301)
INCO	0.0003***	0.00007***	0.00009***	0.00008***
	(0.0001)	(0.00001)	(0.00001)	(0.00001)
Year dummies	Yes	Yes	Yes	Yes
F	312.96***	80.61***	10,956.50***	41.52***
z ₁	0.000	0.000	0.000	0.000
z_2	0.000	0.000	0.000	0.000
m ₂	1.46	1.38	1.49	1.39
Sargan	0.143	0.184	0.242	0.126

*, **, *** Significant at 10 %, 5 % and 1 %, respectively. Standard errors in brackets.

VARIABLES: VARIABLES: LEVER (leverage ratio); DIROWN (percentage of shares owned by the directors);DIROWN2 (the square of the DIROWN variable); CEOWN (percentage of shares owned by the CEO); CEOWN2 (the square of the CEOWN variable); FSIZE (firm size); LFAGE (logarithm of the firm age); INTAS (intangible assets to total assets); DEBCOS (debt cost); ROA (return on assets); INCO (interest coverage ratio); F statistic (test of combined significance); Hausman (Hausman specification test), z_1 and z_2 , are two Wald tests of the joint significance of the reported coefficients and the joint significance of the time dummy variables, respectively (asymptotically distributed as λ^2 under the null hypothesis of no relationship, probability is shown); m_2 is a second-order serial correlation test using residuals in first differences, asymptotically distributed as N(0,1) under the null hypothesis of no serial correlation; Sargan is a test of the over-identifying restrictions, asymptotically distributed as λ^2 under the null hypothesis of no correlation between the instruments and the error term, de grees of freedom in parentheses, probability is shown.

evidence in a country with a high concentration of business ownership in small and medium-sized enterprises.

Table 4 presents all the variables together. This table confirms the previous results, namely a negative effect of the proportion of female directors (PWOMEN), duality (DUAL), and board size (LNDIR), on firm debt (LEVER), and a positive effect of director and CEO ownership (DIROWN and CEOWN). The sign of the control variables is the same as above.

Considering the results obtained in this study of the effect of the share ownership of directors and CEO on indebtedness, we set out to analyze whether the presence of female directors has a moderating effect (Baron & Kenny, 1986).

In this sense, the greater risk aversion that women can provide and the greater heterogeneity that they bring to the board of directors can cause even companies with the highest shareholding concentration to reduce their exposure to financial risk, limiting the power of those who own shares. In other words, we analyze whether the presence of women on boards of directors serves to offset the power of shareholder concentration and thus reduce the debt ratio.

Following the previous arguments, we study the moderating effect of

Table 4

System GMM Estimation of the influence of female directors (PWOMEN), CEO duality (DUAL), board size (LNDIR), the percentage of shares owned by directors (DIROWN) and the percentage of shares owned by CEO (CEOWN) on leverage ratio (LEVER).

Variable	Model 1	Model 2	Model 3	Model 4
Constant	0.077**	-0.154	0.088	-0.330*
	(0.0314)	(0.1709)	(0.0573)	(0.171)
PWOMEN	-0.043***	-0.086**	-0.174***	-0.058*
	(0.0065)	(0.0318)	(0.0244)	(0.0306)
DUAL	-0.040***	-0.026**	-0.058***	-0.035**
	(0.0033)	(0.134)	(0.0042)	(0.0144)
LNDIR	-0.022^{***}	-0.011*	-0.017***	-0.022^{**}
	(0.0017)	(0.0067)	(0.0029)	(0.0102)
DIROWN	0.001***	0.002*		
	(0.00004)	(0.0013)		
DIROWN2		-0.00001		
		(0.00001)		
CEOWN			0.001***	0.002*
			(0.00008)	(0.0014)
CEOWN2				-0.00002
				(0.00001)
FSIZE	0.053***	0.067***	0.055***	0.079***
	(0.0019)	(0.0108)	(0.0036)	(0.0109)
LFAGE	-0.080***	-0.080***	-0.079***	-0.076***
	(0.0009)	(0.0077)	(0.0012)	(0.0077)
INTAS	-0.099***	-0.114***	-0.153***	-0.082^{**}
	(0.011)	(0.0319)	(0.0174)	(0.0361)
DEBCOS	-0.514***	-0.495***	-0.528***	-0.446***
	(0.0257)	(0.114)	(0.0525)	(0.1044)
ROA	-0.051***	-0.269***	-0.374***	-0.282^{***}
	(0.0257)	(0.0693)	(0.0387)	(0.0757)
INCO	0.0001***	0.00008	0.0003***	0.0002
	(0.00001)	(0.0001)	(0.0001)	(0.0001)
Year dummies	Yes	Yes	Yes	Yes
F	772.18***	28.09***	474.79***	28.17***
z_1	0.000	0.000	0.000	0.000
z ₂	0.000	0.000	0.000	0.000
m ₂	1.11	1.37	1.34	1.59
Sargan	0.275	0.171	0.198	0.174

*, **, *** Significant at 10 %, 5 % and 1 %, respectively. Standard errors in brackets.

VARIABLES: LEVER (leverage ratio); PWOMEN (percentage of female directors on the board); DUAL (dummy variable that takes a value of 1 when the chairman of the board and CEO are not the same person, and zero otherwise): LNDIR (the logarithm of the number of directors on the board); DIROWN (percentage of shares owned by the directors); DIROWN2 (the square of the DIROWN variable); CEOWN (percentage of shares owned by the CEO); CEOWN2 (the square of the CEOWN variable); FSIZE (firm size); LFAGE (logarithm of the firm age); INTAS (intangible assets to total assets); DEBCOS (debt cost); ROA (return on assets); INCO (interest coverage ratio); F statistic (test of combined significance); Hausman (Hausman specification test), z1 and z2, are two Wald tests of the joint significance of the reported coefficients and the joint significance of the time dummy variables, respectively (asymptotically distributed as λ^2 under the null hypothesis of no relationship, probability is shown); m₂ is a second-order serial correlation test using residuals in first differences, asymptotically distributed as N(0,1) under the null hypothesis of no serial correlation; Sargan is a test of the over-identifying restrictions, asymptotically distributed as λ^2 under the null hypothesis of no correlation between the instruments and the error term, degrees of freedom in parentheses, probability is shown.

adding women to the board on reducing the leverage of SMEs where more shares are in the hands of directors and the CEO. In Table 5, we observe that more female directors reduce the indebtedness of firms where directors (Model 1) and the CEO (Model 2) own more of the shares. So, women on the board help to reduce the level of firm debt.

The interactive term PWOMENxDIREOWN has significant and negative impact on debt ratio (β =-0.0007, *p*<0.01), meaning that if the percentage of female directors and directors' ownership increase by 1 %, the firm financial risk in terms of debt ratio will decrease by 0.07 %. This can be interpreted as meaning that firms with high director ownership tend to have heterogeneous gender diverse boards which decrease the

Table 5

System GMM Estimation of the interrelationship of the percentage of women on the board (PWOMEN) with the percentage of shares owned by directors (DIR-OWN) and the percentage of shares owned by CEO (CEOWN) on leverage ratio (LEVER).

Constant 0.062^{**} 0.064^{**} (0.0315) (0.0321) PWOMEN -0.019^{**} -0.019^{**} (0.0087) (0.0082) DUAL -0.038^{***} -0.050^{***} (0.0033) (0.0032) LNDIR -0.021^{***} -0.008^{***} (0.0017) (0.0022) DIROWN 0.008^{***} (0.0005) PWOMENxDIROWN -0.0007^{***} (0.0009) CEOWN -0.007^{***} (0.0003) FSIZE 0.054^{***} 0.052^{***} (0.0019) (0.0020) $IFAGE$ -0.089^{***} (0.0019) (0.0020) $IFAGE$ -0.089^{***} -0.103^{***} (0.0019) (0.0009) (0.0009) (0.0009) INTAS -0.089^{***} -0.013^{***} (0.0169) DEBCOS -0.513^{***} -0.056^{***} (0.0163) (0.0163) INCO (0.0001^{**}) $(0.0001^{***}$ $(0.0001^{***}$ $(0.0001^{***}$ <t< th=""><th>Variable</th><th>Model 1</th><th>Model 2</th></t<>	Variable	Model 1	Model 2
PWOMEN -0.019^{**} -0.019^{**} DUAL -0.038^{***} -0.050^{***} (0.0033) (0.0032) LNDIR -0.021^{***} (0.0032) LNDIR -0.021^{***} (0.0032) DIROWN 0.001^{**} (0.0022) DIROWN 0.0008^{***} (0.0023) DIROWN -0.0007^{***} (0.0005) PWOMENxDIROWN -0.0007^{***} (0.0009) CEOWN -0.007^{***} (0.0009) PWOMENXCEOWN -0.051^{***} (0.0003) FSIZE 0.054^{***} (0.0020) LFAGE -0.080^{***} -0.079^{***} (0.0009) (0.0009) (0.0020) INTAS -0.089^{***} -0.103^{***} (0.0115) (0.01169) $0.0263)$ POEOS -0.513^{***} -0.056^{***} (0.0258) (0.0139) 0.0001^{***} INCO $(0.0001)^{***}$ $(0.0001)^{***}$ $Vear dummies$ Yes Yes<	Constant	0.062**	0.064**
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.0315)	(0.0321)
$\begin{array}{cccccccc} \text{DUAL} & -0.038^{***} & -0.050^{***} \\ (0.0033) & (0.0032) \\ \text{LNDIR} & -0.021^{***} & -0.008^{***} \\ (0.0017) & (0.0022) \\ \text{DIROWN} & 0.0008^{***} \\ (0.00005) \\ \text{PWOMENXDIROWN} & -0.0007^{***} \\ (0.0001) & & & & & & & & & & & & & & & & & & &$	PWOMEN	-0.019**	-0.019**
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.0087)	(0.0082)
$\begin{array}{cccccccc} \text{LNDIR} & -0.021^{***} & -0.008^{***} \\ & (0.0017) & (0.0022) \\ \text{DIROWN} & 0.0008^{***} \\ & (0.00005) \\ \text{PWOMENxDIROWN} & -0.0007^{***} \\ & (0.0001) \\ \hline \\ \text{CEOWN} & & 0.001^{***} \\ & (0.0009) \\ \text{PWOMENXCEOWN} & & -0.007^{**} \\ & (0.0003) \\ \text{FSIZE} & 0.054^{***} & (0.0019) \\ & (0.0020) \\ \text{LFAGE} & -0.080^{***} & -0.079^{***} \\ & (0.0019) & (0.0020) \\ \\ \text{LFAGE} & -0.089^{***} & -0.079^{***} \\ & (0.0009) & (0.0009) \\ \text{INTAS} & -0.089^{***} & -0.103^{***} \\ & (0.0115) & (0.01169) \\ \\ \text{DEBCOS} & -0.513^{***} & -0.056^{***} \\ & (0.0258) & (0.0258) \\ \text{ROA} & -0.051^{***} & -0.056^{***} \\ & (0.0136) & (0.0139) \\ \\ \text{INCO} & 0.001^{***} & 0.001^{***} \\ & (0.0001) & (0.0001) \\ \\ \text{Year dummies} & \text{Yes} & \text{Yes} \\ \text{F} & 727.11^{***} & 699.42^{***} \\ z_1 & 0.000 & 0.000 \\ m_2 & 1.04 & 1.08 \\ \end{array}$	DUAL	-0.038***	-0.050***
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.0033)	(0.0032)
$\begin{array}{cccc} \text{DIROWN} & 0.0008^{***} & & & & & & & & & & & & & & & & & &$	LNDIR	-0.021***	-0.008***
$\begin{array}{c} (0.00005) \\ -0.0007^{***} \\ (0.0001) \\ \hline \\ CEOWN \\ \hline \\ CEOWN \\ \hline \\ (0.0002) \\ \hline \\ CEOWN \\ \hline \\ (0.0003) \\ FSIZE \\ (0.0019) \\ (0.0020) \\ \hline \\ FSIZE \\ (0.0019) \\ (0.0020) \\ (0.0020) \\ \hline \\ LFAGE \\ (0.0019) \\ (0.0009) \\ (0.0009) \\ (0.0009) \\ INTAS \\ (0.0019) \\ (0.0009) \\ (0.0009) \\ (0.0009) \\ INTAS \\ (0.0009) \\ (0.0009) \\ INTAS \\ (0.0015) \\ (0.0115) \\ (0.01169) \\ OLEBCOS \\ (0.0115) \\ (0.01169) \\ OLEBCOS \\ (0.0258) \\ (0.0258) \\ (0.0258) \\ (0.0258) \\ (0.0139) \\ INCO \\ (0.0001) \\ (0.0001) \\ (0.0001) \\ INCO \\ (0.0001) \\ (0.0001) \\ (0.0001) \\ Vear dummies \\ Fe \\ Faccocccccccccccccccccccccccccccccccccc$		(0.0017)	(0.0022)
$\begin{array}{cccc} \mbox{PWOMENxDIROWN} & -0.0007^{***} & (0.0001) & 0.001^{***} & (0.0009) & 0.0007^{***} & (0.0009) & 0.0007^{**} & (0.0003) & 0.0003 & 0.0003 & 0.0003 & 0.0003 & 0.0003 & 0.0003 & 0.0003 & 0.0003 & 0.0003 & 0.0003 & 0.0003 & 0.0009 & 0.0003 & 0.0001 & 0.00001 & 0.0001 & 0$	DIROWN	0.0008***	
$\begin{array}{cccc} (0.0001) & & & & & & & & & & & & & & & & & & &$		(0.00005)	
$\begin{array}{c c} {\sf CEOWN} & & 0.001^{***} & (0.0009) \\ & & & & & & & & & & & & & & & & & & $	PWOMENxDIROWN	-0.0007***	
$\begin{array}{llllllllllllllllllllllllllllllllllll$		(0.0001)	
$\begin{array}{cccc} & & & & & & & & & & & & & & & & & $	CEOWN		0.001***
$\begin{array}{llllllllllllllllllllllllllllllllllll$			(0.00009)
$\begin{array}{llllllllllllllllllllllllllllllllllll$	PWOMENxCEOWN		-0.0007**
$\begin{array}{llllllllllllllllllllllllllllllllllll$			(0.0003)
$\begin{array}{cccccc} {\rm LFAGE} & -0.080^{***} & -0.079^{***} \\ & (0.0009) & (0.0009) \\ {\rm INTAS} & -0.089^{***} & -0.103^{***} \\ & (0.0115) & (0.01169) \\ {\rm DEBCOS} & -0.513^{***} & -0.497^{***} \\ & (0.0258) & (0.0263) \\ {\rm ROA} & -0.051^{***} & -0.056^{***} \\ & (0.0136) & (0.0139) \\ {\rm INCO} & 0.0001^{***} & 0.0001^{***} \\ & (0.00001) & (0.00001) \\ {\rm Year \ dummies} & {\rm Yes} & {\rm Yes} \\ {\rm F} & 727.11^{***} & 69.42^{***} \\ {\rm z}_1 & 0.000 & 0.000 \\ {\rm z}_2 & 0.000 & 0.000 \\ {\rm m}_2 & 1.04 & 1.08 \\ \end{array}$	FSIZE	0.054***	0.052***
$\begin{array}{cccc} (0.0009) & (0.0009) \\ \text{INTAS} & -0.089^{***} & -0.103^{***} \\ (0.0115) & (0.01169) \\ \text{DEBCOS} & -0.513^{***} & -0.497^{***} \\ (0.0258) & (0.0258) \\ \text{ROA} & -0.051^{***} & -0.056^{***} \\ (0.0136) & (0.0139) \\ \text{INCO} & 0.0001^{***} & 0.0001^{***} \\ (0.00001) & (0.0001) \\ \text{Year dummies} & \text{Yes} & \text{Yes} \\ \text{F} & 727.11^{***} & 699.42^{***} \\ z_1 & 0.000 & 0.000 \\ z_2 & 0.000 & 0.000 \\ m_2 & 1.04 & 1.08 \\ \end{array}$		(0.0019)	(0.0020)
$\begin{array}{ccccc} \text{INTAS} & -0.089^{***} & -0.103^{***} \\ & (0.0115) & (0.01169) \\ \text{DEBCOS} & -0.513^{***} & -0.497^{***} \\ & (0.0258) & (0.0263) \\ \text{ROA} & -0.051^{***} & -0.056^{***} \\ & (0.0136) & (0.0139) \\ \text{INCO} & 0.0001^{***} & 0.0001^{***} \\ & (0.00001) & (0.00001) \\ \text{Year dummies} & \text{Yes} & \text{Yes} \\ \text{F} & 727.11^{***} & 699.42^{***} \\ \text{z}_1 & 0.000 & 0.000 \\ \text{z}_2 & 0.000 & 0.000 \\ \text{m}_2 & 1.04 & 1.08 \\ \end{array}$	LFAGE	-0.080***	-0.079***
$\begin{array}{cccc} (0.0115) & (0.01169) \\ 0.0258) & -0.513^{***} & -0.497^{***} \\ (0.0258) & (0.0263) \\ ROA & -0.051^{***} & -0.056^{***} \\ (0.0136) & (0.0139) \\ INCO & 0.0001^{***} & 0.0001^{***} \\ (0.00001) & (0.00001) \\ Year dummies & Yes & Yes \\ F & 727.11^{***} & 699.42^{***} \\ z_1 & 0.000 & 0.000 \\ z_2 & 0.000 & 0.000 \\ m_2 & 1.04 & 1.08 \\ \end{array}$		(0.0009)	(0.0009)
$\begin{array}{cccc} \text{DEBCOS} & -0.513^{***} & -0.497^{***} \\ (0.0258) & (0.0263) \\ \text{ROA} & -0.051^{***} & -0.056^{***} \\ (0.0136) & (0.0139) \\ \text{INCO} & 0.0001^{***} & 0.0001^{***} \\ (0.00001) & (0.00001) \\ \text{Year dummies} & \text{Yes} & \text{Yes} \\ \text{F} & 727.11^{***} & 699.42^{***} \\ \text{z}_1 & 0.000 & 0.000 \\ \text{z}_2 & 0.000 & 0.000 \\ \text{m}_2 & 1.04 & 1.08 \\ \end{array}$	INTAS	-0.089***	-0.103^{***}
$\begin{array}{cccc} (0.0258) & (0.0263) \\ \text{ROA} & -0.051^{***} & -0.056^{***} \\ (0.0136) & (0.0139) \\ \text{INCO} & 0.0001^{***} & 0.0001^{***} \\ (0.00001) & (0.00001) \\ \text{Year dummies} & \text{Yes} & \text{Yes} \\ \text{F} & 727.11^{***} & 69.42^{***} \\ z_1 & 0.000 & 0.000 \\ z_2 & 0.000 & 0.000 \\ m_2 & 1.04 & 1.08 \\ \end{array}$		(0.0115)	(0.01169)
$\begin{array}{cccc} ROA & -0.051^{***} & -0.056^{***} \\ & (0.0136) & (0.0139) \\ INCO & 0.0001^{***} & 0.0001^{***} \\ & (0.00001) & (0.00001) \\ Year dummies & Yes & Yes \\ F & 727.11^{***} & 699.42^{***} \\ z_1 & 0.000 & 0.000 \\ z_2 & 0.000 & 0.000 \\ m_2 & 1.04 & 1.08 \\ \end{array}$	DEBCOS	-0.513^{***}	-0.497***
$\begin{array}{cccc} (0.0136) & (0.0139) \\ \text{INCO} & 0.0001^{***} & 0.0001^{***} \\ (0.00001) & (0.00001) \\ \text{Year dummies} & \text{Yes} & \text{Yes} \\ \text{F} & 727.11^{***} & 69.42^{***} \\ z_1 & 0.000 & 0.000 \\ z_2 & 0.000 & 0.000 \\ m_2 & 1.04 & 1.08 \\ \end{array}$		(0.0258)	(0.0263)
$\begin{array}{ccccc} \text{INCO} & 0.0001^{***} & 0.0001^{***} \\ & (0.00001) & (0.00001) \\ \text{Year dummies} & \text{Yes} & \text{Yes} \\ \text{F} & 727.11^{***} & 699.42^{***} \\ \text{z}_1 & 0.000 & 0.000 \\ \text{z}_2 & 0.000 & 0.000 \\ \text{m}_2 & 1.04 & 1.08 \\ \end{array}$	ROA	-0.051***	-0.056***
$\begin{array}{cccc} (0.00001) & (0.00001) \\ \mbox{Year dummies} & Yes & Yes \\ \mbox{F} & 727.11^{***} & 699.42^{***} \\ \mbox{z}_1 & 0.000 & 0.000 \\ \mbox{z}_2 & 0.000 & 0.000 \\ \mbox{m}_2 & 1.04 & 1.08 \\ \end{array}$		(0.0136)	(0.0139)
Year dummies Yes Yes F 727.11^{***} 699.42^{***} z_1 0.000 0.000 z_2 0.000 0.000 m_2 1.04 1.08	INCO	0.0001***	0.0001***
		(0.00001)	(0.00001)
$ \begin{array}{cccc} z_1 & 0.000 & 0.000 \\ z_2 & 0.000 & 0.000 \\ m_2 & 1.04 & 1.08 \end{array} $	Year dummies	Yes	Yes
$\begin{array}{ccccccc} z_2 & 0.000 & 0.000 \\ m_2 & 1.04 & 1.08 \end{array}$	F	727.11***	699.42***
m ₂ 1.04 1.08	z ₁	0.000	0.000
2	z ₂	0.000	0.000
Sargan 0.281 0.273	m ₂		
	Sargan	0.281	0.273

*, **, *** Significant at 10 %, 5 % and 1 %, respectively. Standard errors in brackets.

VARIABLES: LEVER (leverage ratio); PWOMEN (percentage of female directors on the board); DUAL (dummy variable that takes a value of 1 when the chairman of the board and CEO are not the same person, and zero otherwise); LNDIR (the logarithm of the number of directors on the board); DIROWN (percentage of shares owned by the directors); CEOWN (percentage of shares owned by the CEO); FSIZE (firm size); LFAGE (logarithm of the firm age); INTAS (intangible assets to total assets); DEBCOS (debt cost); ROA (return on assets); INCO (interest coverage ratio); F statistic (test of combined significance); Hausman (Hausman specification test), z1 and z2, are two Wald tests of the joint significance of the reported coefficients and the joint significance of the time dummy variables, respectively (asymptotically distributed as λ^2 under the null hypothesis of no relationship, probability is shown); m2 is a second-order serial correlation test using residuals in first differences, asymptotically distributed as N (0,1) under the null hypothesis of no serial correlation; Sargan is a test of the over-identifying restrictions, asymptotically distributed as λ^2 under the null hypothesis of no correlation between the instruments and the error term, degrees of freedom in parentheses, probability is shown.

firms' financial risk. Also, PWOMENxDIREOWN produces similar results.

So, women on the board can help to reduce the level of firm debt. This finding represents a contribution to the study of corporate governance. We are not aware of any similar analysis before. In contexts of high share ownership, such as the Spanish case, adding women to the board of directors can serve to reduce the financial risk, and therefore the bankruptcy, of companies. The presence of women directors is not only important for ethical reasons. There are also the economic reasons, that it can mitigate the devastating effects of financial crises.

In addition, in Tables 6 and 7, we analyze the interrelation of the proportion of women on boards (PDIR) with the separation of positions

J.F. Martín-Ugedo and A. Mínguez-Vera

Table 6

System GMM Estimation of the interrelationship of the percentage of women on the board (PWOMEN) with separation of charges of CEO and Chairman (DUAL) on leverage ratio (LEVER).

Variable	Model 1	Model 2
Constant	0.071*	0.124***
	(0.0361)	(0.0361)
PWOMEN	-0.056***	-0.046***
	(0.0098)	(0.0098)
DUAL	-0.041***	-0.053***
	(0.0041)	(0.0040)
LNDIR	-0.022***	-0.021***
	(0.0020)	(0.0021)
DIROWN	0.0007***	
	(0.00005)	
CEOWN		0.0007***
		(0.00007)
PWOMENxDUAL	0.015	0.005
	(0.0148)	(0.0149)
FSIZE	0.053***	0.051***
	(0.0022)	(0.0022)
LFAGE	-0.080***	-0.077***
	(0.0011)	(0.0011)
INTAS	-0.098***	-0.127***
	(0.0131)	(0.0131)
DEBCOS	-0.510***	-0.504***
	(0.0296)	(0.0298)
ROA	-0.048***	-0.0428***
	(0.0156)	(0.0157)
INCO	0.0001***	0.0001***
	(0.00001)	(0.00001)
Year dummies	Yes	Yes
F	545.84***	536.68 ***
z ₁	0.000	0.000
z ₂	0.000	0.000
m ₂	1.39	1.24
Sargan	0.174	0.180

*, **, *** Significant at 10 %, 5 % and 1 %, respectively. Standard errors in brackets.

VARIABLES: LEVER (leverage ratio); PWOMEN (percentage of female directors on the board); DUAL (dummy variable that takes a value of 1 when the chairman of the board and CEO are not the same person, and zero otherwise); LNDIR (the logarithm of the number of directors on the board); DIROWN (percentage of shares owned by the directors); CEOWN (percentage of shares owned by the CEO); FSIZE (firm size); LFAGE (logarithm of the firm age); INTAS (intangible assets to total assets); DEBCOS (debt cost); ROA (return on assets); INCO (interest coverage ratio); F statistic (test of combined significance); Hausman (Hausman specification test), z1 and z2, are two Wald tests of the joint significance of the reported coefficients and the joint significance of the time dummy variables, respectively (asymptotically distributed as λ^2 under the null hypothesis of no relationship, probability is shown); m2 is a second-order serial correlation test using residuals in first differences, asymptotically distributed as N (0,1) under the null hypothesis of no serial correlation; Sargan is a test of the over-identifying restrictions, asymptotically distributed as λ^2 under the null hypothesis of no correlation between the instruments and the error term, degrees of freedom in parentheses, probability is shown.

of CEO and Chairman, DUAL (Table 6) and the board size (Table 7).

Table 6 shows a non-significant interaction effect, which means that the presence of women on the boards of Spanish SMEs does not produce a positive effect of separation of positions on corporate debt. This occurs whether we include the CEO ownership (Model 1) or the percentage of shares of the members of the board of directors (Model 2) as the dependent variable. There are no changes in the significance or sign of the other variables.

The result is similar when the interaction between the percentage of women and the size of the board is included (Table 7). That is, there is no increase in debt as a consequence of the joint effect.

Finally, in Tables 8 and 9, we study whether the other characteristics of the board, the duality of positions (Table 8) and the board size (Table 9) serve to mitigate the positive effect of both directors' ownership and CEO ownership.

Table 7

System GMM Estimation of the interrelationship of the percentage of women on the board (PWOMEN) with board size (LNDIR) on leverage ratio (LEVER).

Variable	Model 1	Model 2
Constant	0.132***	0.188***
	(0.0360)	(0.0359)
PWOMEN	-0.044***	-0.038***
	(0.0107)	(0.0108)
DUAL	-0.036***	-0.048***
	(0.0037)	(0.0036)
LNDIR	-0.023***	-0.022^{***}
	(0.0022)	(0.0024)
DIROWN	0.0007***	
	(0.00005)	
CEOWN		0.0006***
		(0.00007)
PWOMENxLNDIR	-0.014	-0. 011
	(0.0121)	(0.0122)
FSIZE	0.050***	0.047***
	(0.0022)	(0.0022)
LFAGE	-0.079***	-0.076***
	(0.0011)	(0. 0011)
INTAS	-0.107***	-0.137***
	(0.0133)	(0.0133)
DEBCOS	-0.512^{***}	-0.506***
	(0.0296)	(0.0298)
ROA	-0.036***	-0.029***
	(0.0156)	(0.0156)
INCO	0.0001***	0.0001***
	(0.00,001)	(0.00001)
Year dummies	Yes	Yes
F	547.23 ***	538.42***
z1	0.000	0.000
z2	0.000	0.000
m2	1.24	1.25
Sargan	0.183	0.183

*, **, *** Significant at 10 %, 5 % and 1 %, respectively. Standard errors in brackets.

VARIABLES: LEVER (leverage ratio); PWOMEN (percentage of female directors on the board); DUAL (dummy variable that takes a value of 1 when the chairman of the board and CEO are not the same person, and zero otherwise); LNDIR (the logarithm of the number of directors on the board); DIROWN (percentage of shares owned by the directors); CEOWN (percentage of shares owned by the CEO); FSIZE (firm size); LFAGE (logarithm of the firm age); INTAS (intangible assets to total assets); DEBCOS (debt cost); ROA (return on assets); INCO (interest coverage ratio); F statistic (test of combined significance); Hausman (Hausman specification test), z1 and z2, are two Wald tests of the joint significance of the reported coefficients and the joint significance of the time dummy variables, respectively (asymptotically distributed as λ^2 under the null hypothesis of no relationship, probability is shown); m2 is a second-order serial correlation test using residuals in first differences, asymptotically distributed as N (0,1) under the null hypothesis of no serial correlation; Sargan is a test of the over-identifying restrictions, asymptotically distributed as λ^2 under the null hypothesis of no correlation between the instruments and the error term, degrees of freedom in parentheses, probability is shown.

In Table 8 it can be seen that the interaction between the DUAL variable and the percentage of shares owned by members of the board of directors (Model 1) and the CEO ownership (Model 2) is positive and significant. Therefore, unlike the percentage of women on the board, the separation of positions of Chairman and CEO does not serve to offset the positive effect that ownership has on corporate debt.

The results are similar in Table 9. A positive interaction relationship is observed between the size of the board and the directors' ownership (Model 1) and the percentage of shares owned by the CEO (Model 2). This result highlights the importance and effectiveness of female directors in reducing debt and, therefore, business risk.

For all the models, the z_1 Wald Test shows the combined significance of the coefficients of the independent variables, while z_2 reports the joint significance of the time dummies. The m_2 statistic shows no secondorder serial correlation in the first-difference residuals in all models. This means that the models are not mis-specified. Sargan's Test shows a

Table 8

System GMM Estimation of the interrelationship of the separation of charges of CEO and Chairman (DUAL) with the percentage of shares owned by directors (DIROWN) and the percentage of shares owned by CEO (CEOWN) on leverage ratio (LEVER).

Variable	Model 1	Model 2
Constant	0.134***	0.180***
	(0.0386)	(0.0386)
PWOMEN	-0.032***	-0.028**
	(0.0088)	(0.0089)
DUAL	-0.047***	-0.061***
	(0.0060)	(0.0052)
LNDIR	-0.025***	-0.023^{***}
	(0.0024)	(0.0027)
DIROWN	0.0007***	
	(0.00009)	
DUALxDIROWN	0.0003***	
	(0.0001)	
CEOWN		0.0003**
		(0.0001)
DUALxCEOWN		0.0007***
		(0.0001)
FSIZE	0.050***	0.048***
	(0.0024)	(0.0024)
LFAGE	-0.078***	-0.075***
	(0.0012)	(0.0011)
INTAS	-0.152^{***}	-0.177***
	(0. 0154)	(0.0155)
DEBCOS	-0.479***	-0.483***
	(0.0324)	(0.0326)
ROA	-0.037***	-0.020***
	(0.0177)	(0.0178)
INCO	0.0001***	0.0001***
	(0.00001)	(0.00001)
Year dummies	Yes	Yes
F	491.10***	481.85***
z ₁	0.000	0.000
z ₂	0.000	0.000
m ₂	1.64	1.59
Sargan	0.201	0.327

*, **, *** Significant at 10 %, 5 % and 1 %, respectively. Standard errors in brackets.

VARIABLES: LEVER (leverage ratio); PWOMEN (percentage of female directors on the board); DUAL (dummy variable that takes a value of 1 when the chairman of the board and CEO are not the same person, and zero otherwise): LNDIR (the logarithm of the number of directors on the board); DIROWN (percentage of shares owned by the directors); CEOWN (percentage of shares owned by the CEO); FSIZE (firm size); LFAGE (logarithm of the firm age); INTAS (intangible assets to total assets); DEBCOS (debt cost); ROA (return on assets); INCO (interest coverage ratio); F statistic (test of combined significance); Hausman (Hausman specification test), z1 and z2, are two Wald tests of the joint significance of the reported coefficients and the joint significance of the time dummy variables, respectively (asymptotically distributed as λ^2 under the null hypothesis of no relationship, probability is shown); m2 is a second-order serial correlation test using residuals in first differences, asymptotically distributed as N (0,1) under the null hypothesis of no serial correlation; Sargan is a test of the over-identifying restrictions, asymptotically distributed as λ^2 under the null hypothesis of no correlation between the instruments and the error term, degrees of freedom in parentheses, probability is shown.

lack of correlation between the error term and the instrument used.

5. Conclusions

Small and Medium Enterprises are extremely important in terms of number, employment and sales. As observed in periods of financial crisis, one of the factors positively influencing their survival is avoiding very high leverage. In this manuscript we examine, from the power perspective, how several board characteristics (gender diversity, duality, board size and insider ownership) influence the firm debt ratio for a sample of Spanish SMEs. Previous studies that have addressed this topic mainly focus on listed firms, and use agency theory in their

Table 9

System GMM Estimation of the interrelationship of the board size (LNDIR) with the percentage of shares owned by directors (DIROWN) and the percentage of shares owned by CEO (CEOWN) on leverage ratio (LEVER).

Variable	Model 1	Model 2
Constant	0.096**	0.181**
	(0.0360)	(0.0667)
PWOMEN	-0. 049***	-0.080***
	(0.0075)	(0.0118)
DUAL	-0.045***	-0.068***
	(0.0036)	(0.0053)
LNDIR	-0.029***	-0.031***
	(0.0023)	(0.0031)
DIROWN	0.0002***	
	(0.00005)	
LNDIRxDIROWN	0.0001**	
	(0.00005)	
CEOWN		0.0006***
		(0.00007)
LNDIRxCEOWN		-0. 0002**
		(0.00008)
FSIZE	0.053***	0.051***
	(0.0022)	(0.0042)
LFAGE	-0.080***	-0.078***
	(0.0011)	(0.0015)
INTAS	-0.103^{***}	-0.175^{***}
	(0.0133)	(0.0209)
DEBCOS	-0.513^{***}	-0.686***
	(0.0296)	(0.0943)
ROA	-0.049***	-0.377***
	(0.0156)	(0.0517)
INCO	0.00009***	0.0002***
	(0.00001)	(0.0002)
Year dummies	Yes	Yes
F	553.76***	321.17***
z ₁	0.000	0.000
Z ₂	0.000	0.000
m ₂	1.66	1.57
Sargan	0.175	0.283

*, **, *** Significant at 10 %, 5 % and 1 %, respectively. Standard errors in brackets.

VARIABLES: LEVER (leverage ratio); PWOMEN (percentage of female directors on the board); DUAL (dummy variable that takes a value of 1 when the chairman of the board and CEO are not the same person, and zero otherwise); LNDIR (the logarithm of the number of directors on the board); DIROWN (percentage of shares owned by the directors); CEOWN (percentage of shares owned by the CEO); FSIZE (firm size); LFAGE (logarithm of the firm age); INTAS (intangible assets to total assets); DEBCOS (debt cost); ROA (return on assets); INCO (interest coverage ratio); F statistic (test of combined significance); Hausman (Hausman specification test), z1 and z2, are two Wald tests of the joint significance of the reported coefficients and the joint significance of the time dummy variables, respectively (asymptotically distributed as λ^2 under the null hypothesis of no relationship, probability is shown); m₂ is a second-order serial correlation test using residuals in first differences, asymptotically distributed as N (0,1) under the null hypothesis of no serial correlation; Sargan is a test of the over-identifying restrictions, asymptotically distributed as λ^2 under the null hypothesis of no correlation between the instruments and the error term, degrees of freedom in parentheses, probability is shown.

analysis. However, in the Spanish context and SMEs, the power perspective is probably a more appropriate framework.

All the results obtained support our hypotheses. We find that more female directors, a lower proportion of shares owned by directors and by the CEO, larger boards and the separation of the roles of CEO and Chairman all lead to a decrease in firm debt. We also find that women directors reduce the level of debt in firms where more shares are in the hands of the CEO and directors. Focusing on this last finding, previous evidence has already indicated that women are more risk averse than men and they are more prone to share the power. However, to our knowledge, there is no previous evidence that shows so clearly how the presence of more women on the board can offset the negative effects of high levels of insider ownership. Furthermore, we find that the other board characteristics examined, separation of positions and size, do not exert a moderating effect on the share ownership of directors and CEOs to reduce corporate debt. Finally, the combination of the proportion of women with the separation of positions and size does not cause an increase in financial risk.

These results are important as they show that the characteristics of the boards of directors are important when it comes to reducing corporate debt and preventing financial crises. The proportion of women on boards of directors is especially decisive, since they not only reduce financial risk but have a moderating effect on insider ownership. Therefore, it is important that legislation is used to increase the proportion of female directors, not only in large companies, but also in SMEs, which are the vast majority of firms in the world economy.

The main limitation of this work is that it is focuses on a single country. Other relevant variables, such as the type of director, have not been included (due to lack of data). For future research, we will try to extend the sample to different countries and legal systems.

Acknowledgements

This work was supported by Spanish Ministry of Science, Innovation and Universities (Spain) under grant RTI2018–093884-B-I00; Fundación Séneca (Spain), under grant 21947/PI/22 and Fundación Cajamurcia. We wish to thank the editor and the anonymous reviewers for their constructive suggestions.

References

- Adams, R. B., Almeida, H., & Ferreira, D. (2005). Powerful CEOs and their impact on corporate performance. *The Review of Financial Studies*, 18(4), 1403–1432. https:// doi.org/10.1093/rfs/hhi030
- Adams, R. B., & Ferreira, D. (2009). Women in the boardroom and their impact on governance and performance. *Journal of Financial Economics*, 94(2), 291–309. https://doi.org/10.1016/j.jfineco.2008.10.007
- Andrade, G., & Kaplan, S. N. (1998). How costly is financial (not economic) distress? Evidence from highly leveraged transactions that became distressed. *The Journal of Finance*, 53(5), 1443–1493. https://doi.org/10.1111/0022-1082.00062
- Arellano, M., & Bond, S. (1991). Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. *The Review of Economic Studies*, 58(2), 277–297. https://doi.org/10.2307/2297968
- Arellano, M., & Bover, O. (1995). Another look at the instrumental variable estimation of error-components models. *Journal of Econometrics*, 68(1), 29–51. https://doi.org/ 10.1016/0304-4076(94)01642-D
- Baixauli-Soler, J. S., & Sanchez-Marin, G. (2015). Executive compensation and corporate governance in Spanish listed firms: A principal-principal perspective. *Review of Managerial Science*, 9(1), 115–140. https://doi.org/10.1007/s11846-014-0122-z
- Bajaj, M., Chan, Y. C., & Dasgupta, S. (1998). The relationship between ownership, financing decisions and firm performance: A signalling model. *International Economic Review*, 39(3), 723–744. https://doi.org/10.2307/2527397
- Banerjee, A., Nordqvist, M., & Hellerstedt, K. (2020). The role of the board chair-A literature review and suggestions for future research. *Corporate Governance: An International Review*, 28(6), 372–405. https://doi.org/10.1111/corg.12350
- Baron, R. M., & Kenny, D. A. (1986). The moderator-mediator variable distinction in social psychological research. Conceptual, strategic, and statistical considerations. *Journal of Personality and Social Psychology*, 51(6), 1173–1182. https://doi.org/ 10.1037/0022-3514.51.6.1173
- Barry, C. B., Mann, S. C., Mihov, V. T., & Rodriguez, M. (2008). Corporate debt issuance and the historical level of interest rates. *Financial Management*, 37(3), 413–430. https://doi.org/10.1111/j.1755-053X.2008.00019.x
- Baxter, N. D. (1967). Leverage, risk of ruin and the cost of capital. The Journal of Finance, 22(3), 395–403. https://doi.org/10.2307/2978892
- Blundell, R., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics*, 87(1), 115–143. https://doi.org/ 10.1016/S0304-4076(98)00009-8
- Brav, O. (2009). Access to capital, capital structure, and the funding of the firm. *The Journal of Finance*, 64(1), 263–308. https://doi.org/10.1111/j.1540-6261.2008.01434.x
- Carter, S. & Shaw, E. (2006). Women's business ownership: Recent research and policy developments. Report to the Small Business Service. Available at https://strathprints .strath.ac.uk/8962/1/SBS_2006_Report_for_BIS.pdf.
- Carter, D. A., Simkins, B. J., & Simpson, W. G. (2003). Corporate governance, board diversity, and firm value. *Financial Review*, 38(1), 33–53. https://doi.org/10.1111/ 1540-6288.00034
- Cashman, Gillan, & Jun. (2012). Going overboard? On busy directors and firm value. Journal of Banking & Finance, 36(12), 3248–3259. https://doi.org/10.1016/j. jbankfin.2012.07.003

- Chaganti, R. (1986). Management in women-owned enterprises. Journal of Small Business Management, 24(4), 18–29. Available at https://typeset.io/papers/management-in -women-owned-enterprises-1ph2vx40i5?citations_has_pdf=true.
- Cheng, S. (2008). Board size and the variability of corporate performance. Journal of Financial Economics, 87(1), 157–176. https://doi.org/10.1016/j.jfineco.2006.10.006
- Core, J. E., Holthausen, R. W., & Larcker, D. F. (1999). Corporate governance, chief executive officer compensation, and firm performance. *Journal of Financial Economics*, 51(3), 371–406. https://doi.org/10.1016/S0304-405X(98)00058-0
- Daily, C. M., & Dalton, D. R. (1994a). Bankruptcy and corporate governance: The impact of board composition and structure. Academy of Management Journal, 37(6), 1603–1617. https://doi.org/10.5465/256801
- Daily, C. M., & Dalton, D. R. (1994b). Corporate governance and the bankrupt firm: An empirical assessment. Strategic Management Journal, 15(8), 643–654. https://doi. org/10.1002/smj.4250150806
- de Enrique Arnau, L., & Pinillos-Costa, M. J (2023). Board of directors and business transformation: A bibliometric analysis. European Journal of Management and Business Economics. https://doi.org/10.1108/EJMBE-10-2022-0338
- Donaldson, L., & Davis, J. H. (1991). Stewardship theory or agency theory: CEO governance and shareholder returns. Australian Journal of Management, 16(1), 49–64. https://doi.org/10.1177/031289629101600103
- Eurostat (2023). Key figures on european business, May. European Commission, Luxemburg. Retrieved from https://ec.europa.eu/eurostat/documents/15216629/1 6825859/KS-04-23-195-EN-N.pdf/3f72cb73-0ba3-bcfe-0a2f-7bbc03ea5361?ver sion=2.0&t=1685428985626. Accessed August 21, 2023.
- Fahlenbrach, R., Low, A., & Stulz, R. M. (2010). Why do firms appoint CEOs as outside directors? *Journal of Financial Economics*, 97(1), 12–32. https://doi.org/10.1016/j. jfineco.2010.01.003
- Fasano, F., Sánchez-Vidal, F. J., & La Rocca, M. (2022). The role of government policies for Italian firms during the COVID-19 crisis. *Finance Research Letters*, 50, Article 103273. https://doi.org/10.1016/j.frl.2022.103273
- Finkelstein, S. (1992). Power in top management teams: Dimensions, measurement, and validation. Academy of Management Journal, 35(3), 505–538. https://doi.org/ 10.5465/256485
- Forbes, D. P., & Milliken, F. J. (1999). Cognition and corporate governance: Understanding boards of directors as strategic decision-making groups. Academy of Management Review, 24(3), 489–505. https://doi.org/10.2307/259138
- García, C. J., & Herrero, B. (2021). Female directors, capital structure and financial distress. Journal of Business Research, 136, 592–601. https://doi.org/10.1016/j. jbusres.2021.07.061
- García-Martín, C. J., & Herrero, B. (2018). Boards of directors: Composition and effects on the performance of the firm. *Economic Research-Ekonomska Istraživanja*, 31(1), 1015–1041. https://doi.org/10.1080/1331677X.2018.1436454
- Gaud, P., Jani, E., Hoesli, M., & Bender, A. (2005). The capital structure of Swiss companies: An empirical analysis using dynamic panel data. *European Financial Management*, *11*(1), 51–69. https://doi.org/10.1111/j.1354-7798.2005.00275.x
 Gorman, M., & Sahlman, W. A. (1989). What do venture capitalists do? *Journal of*
- Business Venturing, 4(4), 231–248. https://doi.org/10.1016/0883-9026(89)90014-1 Grosvold. J., & Brammer, S. (2011). National institutional systems as antecedents of
- female board representation: An empirical study. Corporate Governance: An International Review, 19(2), 116–135. https://doi.org/10.1111/j.1467-8683.2010.00830.x
- Hall, G. C., Hutchinson, P. J., & Michaelas, N. (2004). Determinants of the capital structures of European SMEs. *Journal of Business Finance & Accounting*, 31(5–6), 711–728. https://doi.org/10.1111/j.0306-686X.2004.00554.x
- Halliday, C. S., Paustian-Underdahl, S. C., & Fainshmidt, S. (2021). Women on boards of directors: A meta-analytic examination of the roles of organizational leadership and national context for gender equality. *Journal of Business and Psychology*, 36(2), 173–191. https://doi.org/10.1007/s10869-019-09679-v
- Hermalin, B. E., & Weisbach, M. S. (2003). Boards of directors as an endogenously determined institution: A survey of the economic literature. *Economic Policy Review*, 9(1), 7–26. Available at https://papers.ssrn.com/sol3/papers.cfm?abstra ct_id=233111.
- Hernández-Cánovas, G., Mínguez-Vera, A., & Sánchez-Vidal, J. (2016). Ownership structure and debt as corporate governance mechanisms: An empirical analysis for Spanish SMEs. Journal of Business Economics and Management, 17(6), 960–976. https://doi.org/10.3846/16111699.2013.859171
- Hernández-Nicolás, C. M., Martín-Ugedo, J. F., & Mínguez-Vera, A. (2019). The effect of gender diversity on the board of Spanish agricultural cooperatives on returns and debt: An empirical analysis. Agribusiness: An International Journal, 35, 639–656. https://doi.org/10.1002/agr.21608
- Himmelberg, C. P., Hubbard, R. G., & Palia, D. (1999). Understanding the determinants of managerial ownership and the link between ownership and performance. *Journal* of Financial Economics, 53(3), 353–384. https://doi.org/10.1016/S0304-405X(99) 00025-2

Hofstede, G. (2021). Hofstede Insights. Retrieved from https://www.hofstede-insights.co m/country-comparison-tool?countries=spain. Accessed February 5, 2023.

- International Council of Small Business (2023). The 6th Edition of the ICSB Global MSMEs Report, June 27. Retrieved from https://www.un.org/sites/un2.un.org/files/icsb-global-msmes-report-2023.pdf . Accessed September 2, 2023.
- Jensen, M. C. (1993). The modern industrial revolution, exit, and the failure of internal control systems. *The Journal of Finance*, 48(3), 831–880. https://doi.org/10.1111/ j.1540-6261.1993.tb04022.x
- Jensen, M. C., & Meckling, W. H. (1976). Theory of the firm: Managerial behavior, agency costs and ownership structure. *Journal of Financial Economics*, 3(4), 305–360. https://doi.org/10.1016/0304-405X(76)90026-X

John, K., & Senbet, L. W. (1998). Corporate governance and board effectiveness. Journal of Banking and Finance, 22(4), 371–403. https://doi.org/10.1016/S0378-4266(98) 00005-3

Judge, W. Q., Jr., & Zeithaml, C. P (1992). Institutional and strategic choice perspectives on board involvement in the strategic decision process. Academy of Management Journal, 35(4), 766–794. https://doi.org/10.5465/256315

Kelly, R. M., Hale, M. M., & Burgess, J. (2008). Gender and managerial/leadership styles. Women & Politics, 11(2), 19–39. https://doi.org/10.1300/J014V11N02_02

Khan, W. A., & Vieito, J. P. (2013). CEO gender and firm performance. Journal of Economics and Business, 67, 55–66. https://doi.org/10.1016/j.jeconbus.2013.01.003

Kieschnick, R., & Moussawi, R. (2018). Firm age, corporate governance, and capital structure choices. *Journal of Corporate Finance*, 48, 597–614. https://doi.org/ 10.1016/j.jcorpfin.2017.12.011

La Porta, R., Lopez-de-Silanes, F., & Shleifer, A (1999). Corporate ownership around the world. The Journal of Finance, 54(2), 471–517. https://doi.org/10.1111/0022-1082.00115

La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R (2002). Investor protection and corporate valuation. *The Journal of Finance*, 57(3), 1147–1170. https://doi.org/ 10.1111/1540-6261.00457

Leech, D., & Manjón, M. C. (2002). Corporate governance in Spain (with an application of the power indices approach). *European Journal of Law and Economics*, 13(2), 157–173. https://doi.org/10.1023/A:1013680820398

López-Delgado, P., & Diéguez-Soto, J. (2020). Indebtedness in family-managed firms: The moderating role of female directors on the board. *Review of Managerial Science*, 14(4), 727–762. https://doi.org/10.1007/s11846-018-0307-y

May, D. O. (1995). Do managerial motives influence firm risk reduction strategies? Journal of Finance, 50(4), 1291–1309. https://doi.org/10.1111/j.1540-6261.1995. tb04059.x

Maxfield, S., Shapiro, M., Gupta, V., & Hass, S. (2010). Gender and risk: Women, risk taking and risk aversion. Gender in Management: An International Journal, 25(7), 586–604. https://doi.org/10.1108/17542411011081383

Merchant, K. (2012). How Men and Women Differ: Gender Differences in Communication Styles, Influence Tactics, and Leadership Styles. CMC Senior Theses. Paper 513, Available at https://scholarship.claremont.edu/cmc theses/513.

Michaelas, N., Chittenden, F., & Poutziouris, P. (1999). Financial policy and capital structure choice in UK SMEs: Empirical evidence from company panel data. *Small Business Economics*, 12(2), 113–130. https://doi.org/10.1023/A:1008010724051

Mínguez-Vera, A., & Martín-Úgedo, J. F. (2010). Firm risk and the power of the Chairman and CEO in a civil law country: Evidence from Spain. *The International Journal of Human Resource Management*, 21(3), 371–388. https://doi.org/10.1080/ 09585190903546912

Ministerio de Industria, Comercio y Turismo (2023). Cifras Pyme, January. Retrieved from https://industria.gob.es/es-es/estadisticas/Cifras_PYME/CifrasPYME-ener o2023.pdf. Accessed August 21, 2023.

Mishra, S., & Mohanty, P. (2014). Corporate governance as a value driver for firm performance: Evidence from India. Corporate Governance, 14(2), 265–280. https:// doi.org/10.1108/CG-12-2012-0089

Modigliani, F., & Miller, M.H. (1958). The cost of capital, corporation finance and the theory of investment. The American Economic Review, 48(3), 261–297. Available at http://www.jstor.org/stable/1809766.

Modigliani, F., & Miller, M.H. (1963). Corporate income taxes and the cost of capital: A correction. The American Economic Review, 53(3), 433–443. Available at http:// www.istor.org/stable/2352621.

Morck, R., Shleifer, A., & Vishny, R. W. (1988). Management ownership and market valuation: An empirical analysis. *Journal of Financial Economics*, 20, 293–315. https://doi.org/10.1016/0304-405X(88)90048-7

Moscovici, S., & Zavalloni, M. (1969). The group as a polarizer of attitudes. Journal of Personality and Social Psychology, 12(2), 125. https://doi.org/10.1037/h0027568

- Myers, S. C., & Majluf, N. S. (1984). Corporate financing and investment decisions when firms have information that investors do not have. *Journal of Financial Economics*, 13 (2), 187–221. https://doi.org/10.1016/0304-405X(84)90023-0
- Nakano, M., & Nguyen, P. (2012). Board size and corporate risk taking: Further evidence from Japan. Corporate Governance: An International Review, 20(4), 369–387. https:// doi.org/10.1111/j.1467-8683.2012.00924.x

- Nielsen, S., & Huse, M. (2010). Women directors' contribution to board decision-making and strategic involvement: The role of equality perception. *European Management Review*, 7(1), 16–29. https://doi.org/10.1057/emr.2009.27
- Olsen, R. A., & Cox, C. M. (2001). The influence of gender on the perception and response to investment risk: The case of professional investors. *The Journal of Psychology and Financial Markets*, 2(1), 29–36. https://doi.org/10.1207/S15327760JPFM0201_3

Parsons, C., & Titman, S. (2009). Empirical capital structure: A review. Foundations and Trends in Finance, 3(1), 1–93. https://doi.org/10.1561/0500000018

Palmer, T. B., & Wiseman, R. M. (1999). Decoupling risk taking from income stream uncertainty: A holistic model of risk. *Strategic Management Journal*, 20(11), 1037–1062. https://doi.org/10.1002/(SICI)1097-0266(199911)20:11<1037::AID-SMJ67>3.0.CO:2-2

Pfeffer, J (1981). Power in organizations. Marshfield, MA: Pitman. Available at http s://www.biblio.com/book/power-organizations-j-pfeffer/d/1235958633? placement=morelikethis.

Roberson, Q. M. (2019). Diversity in the workplace: A review, synthesis, and future research agenda. Annual Review of Organizational Psychology and Organizational Behavior, 6, 69–88. https://doi.org/10.1146/annurev-orgpsych-012218-015243

- Robinson, T. R., Henry, E., Pirie, W. L., & Broihahn, M. A. (2015). International financial statement analysis (3rd ed.). New York: John Wiley & Sons. (Chapter 7). Available at https://www.oreilly.com/library/view/international-financial-statement/978111 8999479/.
- Rosener, J. B. (1990). Ways women lead. Harvard Business Review, 68(2), 119–125. https://doi.org/10.1007/978-90-481-9014-0_3
- Sah, R., & Stiglitz, J. (1986). The architecture of economic systems: Hierarchies and polyarchies. *The American Economics Review*, 76(4), 716–727. Available at http:// www.jstor.org/stable/1806069.
- Sah, R. K., & Stiglitz, J. E. (1991). The quality of managers in centralized versus decentralized organizations. *The Quarterly Journal of Economics*, 106(1), 289–295. https://doi.org/10.2307/2937917
- Sánchez-Marín, G., Lucas-Pérez, M. E., Baixauli-Soler, S., Main, B. G., & Mínguez-Vera, A. (2022). Excess executive compensation and corporate governance in the United Kingdom and Spain: A comparative analysis. *Managerial and Decision Economics*, 43 (7), 2817–2837. https://doi.org/10.1002/mde.3564

Sánchez-Vidal, J., Hernández-Robles, M., & Mínguez-Vera, A. (2020). Financial conservatism fosters job creation during economic crises. *Applied Economics*, 52(45), 4913–4926. https://doi.org/10.1080/00036846.2020.1751053

Spanish Good Governance Code (2020). Comisión Nacional del Mercado de Valores, Madrid. Retrieved from https://www.cnmv.es/DocPortal/Publicaciones/CodigoGo v/CBG 2020 ENen.PDF. Accessed July 4, 2023.

- Tajfel, H., & andTurner, J. C. (2004). The social identity theory of intergroup behavior. In J. T. Jost, & J. Sidanius (Eds.), *Political psychology: Key readings* (pp. 276–293). London: Psychology Press. https://doi.org/10.4324/9780203505984-16.
- Titman, S., & Wessels, R. (1988). The determinants of capital structure choice. The Journal of Finance, 43(1), 1–19. https://doi.org/10.1111/j.1540-6261.1988. tb02585.x
- Valls Martínez, M. C., Martín-Cervantes, P. A., & del Mar Miralles-Quirós, M. (2022). Sustainable development and the limits of gender policies on corporate boards in Europe. A comparative analysis between developed and emerging markets. *European Research on Management and Business Economics*, 28(1), Article 100168. https://doi. org/10.1016/j.iedeen.2021.100168

Warner, J. B. (1977). Bankruptcy costs: Some evidence. The Journal of Finance, 32(2), 337–347. https://doi.org/10.2307/2326766

- Watson, W. E., Kumar, K., & Michaelsen, L. K. (1993). Cultural diversity's impact on interaction process and performance: Comparing homogeneous and diverse task groups. Academy of Management Journal, 36(3), 590–602. https://doi.org/10.2307/ 256593
- World Bank. (2012). World development report 2012. Gender equality and development. Washington, D.C.: World Bank. Retrieved from https://openknowledge.worldbank. org/handle/10986/4391 Accessed October 4, 2022.

Yermack, D. (1996). Higher market valuation of companies with a small board of directors. Journal of Financial Economics, 40(2), 185–211. https://doi.org/10.1016/ 0304-405X(95)00844-5

Zarebski, P., & Dimovski, B. (2012). Determinants of capital structure of A-REITS and the global financial crisis. *Pacific Rim Property Research Journal*, 18(1), 3–19. https://doi. org/10.1080/14445921.2012.11104347